EUROPEAN CENTRAL BANK WORKING PAPER SERIES



WORKING PAPER NO. 174

INTERNATIONAL MONETARY POLICY COORDINATION AND FINANCIAL MARKET INTEGRATION



BY ALAN SUTHERLAND

September 2002

EUROPEAN CENTRAL BANK

WORKING PAPER SERIES





WORKING PAPER NO. 174

INTERNATIONAL MONETARY POLICY COORDINATION AND FINANCIAL MARKET INTEGRATION'

BY ALAN SUTHERLAND²

September 2002

- 1 This research was supported by the ESRC Evolving Macroeconomy Programme grant number L138251046.
- The opinions expressed herein are those of the authors and do not necessarily represent those of the European Central Bank.
- 2 Department of Economics, University of St Andrews, Fife, KY16 9 AL, UK. Email: ajs10@st-and.ac.uk Web: www.st.-and.ac.uk/~ajs10/ home.html

International Research Forum on Monetary Policy

The International Research Forum on Monetary Policy is a new series of conferences organised and sponsored by the European Central Bank, the Board of Governors of the Federal Reserve System, the Center for German and European Studies at Georgetown University and the Center for Financial Studies at Goethe University. The conferences aim to provide opportunities to discuss new research on theoretical and empirical aspects of monetary policy that are relevant from a global perspective.

The Forum will meet regularly at locations alternating between the euro area and the United States.

The first conference in this new series took place at the European Central Bank on 5 and 6 July 2002. The issues, numbers 174 to 181 of the European Central Bank Working Paper Series contain the papers presented on that occasion. The programme of the conference and the invited discussions can be accessed at <u>http://</u> www.ecb.int/events/conf/intforum/.

© European Central Bank, 2002

Address	Kaiserstrasse 29
	D-60311 Frankfurt am Main
	Germany
Postal address	Postfach 16 03 19
	D-60066 Frankfurt am Main
	Germany
Telephone	+49 69 1344 0
Internet	http://www.ecb.int
Fax	+49 69 1344 6000
Telex	411 144 ecb d

All rights reserved.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged. The views expressed in this paper do not necessarily reflect those of the European Central Bank.

ISSN 1561-0810

Contents

Abst	ract	4
Non	-technical summary	5
I	Introduction	7
2	The model2.1Market structure2.2Preferences2.3Price indices2.4Consumption choices2.5Optimal price setting2.6Home and foreign shocks2.7Money demand and supply2.8Financial markets and risk sharing	10 10 11 12 12 13 13
3	Welfare	15
4	 The welfare gains from policy coordination: the logarithmic utility case 4.1 Monetary policy, the exchange rate and output 4.2 Financial autarky 4.3 Risk sharing 	16 16 18 22
5	 Risk aversion and the welfare gains from policy coordination 5.1 Financial autarky 5.2 Risk sharing 5.3 Symmetric shocks and the gains from policy coordination 	25 25 27 28
6	The welfare gains from risk sharing	29
7	The optimality of price targeting	30
8	Conclusion	31
Арре	endix Optimal price setting Model solution An alternative risk sharing structure	32 32 33 35
Refe	rences	36
Euro	pean Central Bank working paper series	40

Abstract

The welfare gains from international coordination of monetary policy are analysed in a two-country model with sticky prices. The gains from coordination are compared under two alternative structures for financial markets: financial autarky and risk sharing. The welfare gains from coordination are found to be largest when there is risk sharing and the elasticity of substitution between home and foreign goods is greater than unity. When there is no risk sharing the gains to coordination are almost zero. It is also shown that the welfare gain from risk sharing can be negative when monetary policy is uncoordinated. Keywords: monetary policy coordination, financial integration, risk sharing.

JEL: E52, E58, F42

Non-technical summary

What are the gains from international coordination of monetary policy? This is a long-standing question in international macroeconomics which was the subject of an extensive literature in the 1980's. More recently attention has returned to the topic following the development of new approaches to analysing the welfare effects of monetary policy in closed and open economies. The 'new open economy macroeconomics literature' emphasises the use of microfounded models and utility-based welfare measures. Obstfeld and Rogoff (2002) analyse the welfare gains from monetary policy coordination in a model of this type. They show that welfare gains do exist but are likely to be very small, both in absolute terms and relative terms (when compared to the welfare costs of business cycle fluctuations).

But the model used by Obstfeld and Rogoff (2002) is special in two respects which are likely to have important implications for the welfare gains from policy coordination. Firstly, the elasticity of substitution between home and foreign goods is restricted to unity. This elasticity determines the extent to which exchange rate changes cause changes in demand for goods from different countries. It is therefore an important determinant of the spillover effect of monetary policy from one country to another. Secondly, Obstfeld and Rogoff assume that international financial markets do not exist. The trade balance is therefore forced into exact balance in all states of the world. Again this removes a potential source of international spillover effects of monetary policy.

The assumption that financial markets do not exist is to some extent less extreme than it may seem at first. When the elasticity of substitution between home and foreign goods is unity the trade balance is always close to balance in any case. The structure of international financial markets is therefore largely irrelevant.

The structure of financial markets does however become much more important when the elasticity of substitution between home and foreign goods differs from unity. In this case large trade imbalances are possible so the structure of financial markets will have an important influence on the behaviour of the exchange rate and the consequential spillover effects of monetary policy. Benigno and Benigno (2001a) analyse a model similar to the Obstfeld and Rogoff (2002) model which allows for a non-unit elasticity of substitution between home and foreign goods and which assumes a financial structure which permits full international consumption risk sharing. They show that the gains from coordination depend on the degree of elasticity of substitution, but in general Benigno and Benigno are not able to solve explicitly for welfare or quantify the gains from coordination.

A constraint that has hitherto hampered progress on this issue is the fact that it is not possible to obtain an explicit exact solution for welfare when the elasticity of substitution between home and foreign goods differs from unity. This paper adopts an approximation technique to overcome this problem. Second-order accurate solutions for welfare are obtained for the general case where the elasticity of substitution differs from unity. This allows explicit solutions for the coordinated and non-coordinated policy rules to be obtained and explicit expressions for the welfare yielded by coordinated and non-coordinated policy to be derived. It is therefore possible to trace the spillover effects which give rise to gains from policy coordination and it is possible to quantify these gains.

The model is used to investigate the implications of the elasticity of substitution for the gains from policy coordination. The implications of financial market structure are also analysed. The gains from coordination that arise when there is no financial market are compared to the gains that arise when there is international risk sharing.

In the case where there is no financial market it is found that a non-unit elasticity of substitution can indeed give rise to gains from coordination. But, as in the cases analysed by Obstfeld and Rogoff (2002), these gains are quantitatively very small. But in the risk-sharing case it is found that the gains from coordination can be much higher. The existence of financial markets creates additional spillover effects which greatly increase the gains from policy coordination. Quantitatively these gains can be quite large in both absolute and relative terms.

Another way to look at the results presented in this paper is to consider the welfare gains from risk sharing. It is found that when monetary policy is coordinated the welfare level achieved in the risk-sharing case is unambiguously higher than the welfare level in the case where there is no financial market. But when monetary policy is not coordinated the answer is very different. In this case the gains from risk sharing are offset by the additional monetary policy spillover effects generated by the existence of financial markets. These spillover effects can be so strong that, for some parameter combinations, the risk-sharing case yields lower welfare than the case where there are no financial markets.

1 Introduction

What are the gains from international coordination of monetary policy? This is a long-standing question in international macroeconomics which was the subject of an extensive literature in the 1980's (see for instance Canzoneri and Henderson (1991), Currie and Levine (1984), Miller and Salmon (1984), Oudiz and Sachs (1984) and Rogoff (1985)). More recently attention has returned to the topic following the development of new approaches to analysing the welfare effects of monetary policy in closed and open economies. The 'new open economy macroeconomics literature' emphasises the use of microfounded models and utility-based welfare measures.¹ Obstfeld and Rogoff (2002) analyse the welfare gains from monetary policy coordination in a model of this type. They show that welfare gains do exist but are likely to be very small, both in absolute terms and relative terms (when compared to the welfare costs of business cycle fluctuations).

But the model used by Obstfeld and Rogoff (2002) is special in two respects which are likely to have important implications for the welfare gains from policy coordination. Firstly, the elasticity of substitution between home and foreign goods is restricted to unity. This parameter determines the strength of the expenditure switching effect of exchange rate changes and is therefore an important determinant of the spillover effect of monetary policy. Secondly, Obstfeld and Rogoff assume that international financial markets do not exist. The trade balance is therefore forced into exact balance in all states of the world. Again this removes a potential source of international spillover effects of monetary policy.

The assumption of financial autarky is to some extent less extreme than it may seem at first. It is a well-known result that when the elasticity of substitution between home and foreign goods is unity and utility is logarithmic in consumption, the trade balance is always in balance in any case.² The structure of international financial markets is therefore irrelevant. It is only in the cases where Obstfeld and Rogoff consider non-logarithmic utility that the structure of financial markets becomes relevant.

The structure of financial markets does however become much more important when the elasticity of substitution between home and foreign goods differs from unity. In this case the trade balance does not automatically balance in all states of the world so the structure of financial markets will have an important influence on

¹See for instance Corsetti and Pesenti (2001a), Devereux and Engel (1998, 2000) and Obstfeld and Rogoff (1995, 1998, 2002). A recent survey of the literature is provided by Lane (2001).

²If all goods are traded then this result holds even when utility is not logarithmic in consumption.

the behaviour of the exchange rate and the consequential spillover effects of monetary policy. Benigno and Benigno (2001a) analyse a model similar to the Obstfeld and Rogoff (2002) model which allows for a non-unit elasticity of substitution between home and foreign goods and which assumes a financial structure which permits full international consumption risk sharing. They show that the gains from coordination depend on the degree of elasticity of substitution, but in general Benigno and Benigno are not able to solve explicitly for welfare or quantify the gains from coordination.

A constraint that has hitherto hampered progress on this issue is the fact that it is not possible to obtain an explicit exact solution for welfare when the elasticity of substitution between home and foreign goods differs from unity. This paper adopts a second-order approximation technique to overcome this problem. Second-order accurate solutions for welfare are obtained for the general case where the elasticity of substitution differs from unity. This allows explicit solutions for the coordinated and non-coordinated policy rules to be obtained and explicit expressions for the welfare yielded by coordinated and non-coordinated policy to be derived. It is therefore possible to trace the spillover effects which give rise to gains from policy coordination and it is possible to quantify these gains.

The model is used to investigate the implications of the elasticity of substitution for the gains from policy coordination. The implications of financial market structure are also analysed. The gains from coordination that arise when there is no financial market are compared to the gains that arise when there is international risk sharing.

In the financial autarky case it is found that a non-unit elasticity of substitution can indeed give rise to gains from coordination. But, as in the cases analysed by Obstfeld and Rogoff (2002), these gains are quantitatively very small. The spillover effects generated by the expenditure switching effect therefore seem to be unimportant when financial markets do not exist. But in the risk-sharing case it is found that the gains from coordination can be much higher. The existence of financial markets creates additional spillover effects which greatly increase the gains from policy coordination. Quantitatively these gains can be quite large in both absolute and relative terms.

Another way to look at the results presented in this paper is to consider the welfare gains from risk sharing. It is found that when monetary policy is coordinated the welfare level achieved in the risk-sharing case is unambiguously higher than the welfare level in the autarky case. But when monetary policy is not coordinated the answer is very different. In this case the gains from risk sharing are offset by the additional monetary policy spillover effects generated by the existence of financial markets. These spillover effects can be so strong that, for some parameter combinations, autarky yields higher welfare than risk sharing.

There have been a number of other contributions to the recent literature which are relevant to the subject of this paper. Corsetti and Pesenti (2001b) analyse the gains from monetary policy coordination when there is incomplete pass-through from exchange rate changes to local currency prices. They show that there are gains to coordination when there is incomplete pass-through even when the elasticity of substitution between home and foreign goods is unity. Clarida, Gali and Gertler (2001) analyse the welfare effects of monetary policy coordination in a model where there are non-optimal 'cost-push' shocks. Again they show that gains from coordination can arise even when the elasticity of substitution between home and foreign goods is unity. Benigno and Benigno (2001b) also consider cost-push shocks but do not consider non-coordinated policy. They show that the optimal coordinated policy can be sustained by each individual monetary authority pursuing a policy of flexible inflation targeting (when 'flexible inflation targeting' is of the form suggested by Svensson (1999)). Benigno (2001) analyses the implications of financial market structure for optimal coordinated policy. He compares an incomplete financial market (where trade is restricted to non-contingent bonds) with full risk sharing. Devereux (2001) also considers the implications of financial market structure. He compares the welfare implications of fixed and flexible exchange rates in the cases of financial autarky and full risk sharing. Tille (1999) analyses the role of the elasticity of substitution between home and foreign goods in the international transmission of shocks. He shows, using a deterministic model, that monetary policy can have a positive or a negative impact on foreign welfare depending on the degree of international substitutability.

This paper proceeds as follows. Section 2 presents the model. Section 3 briefly discusses the measurement of welfare. Section 4 analyses the gains from policy coordination in the special case where utility is logarithmic in consumption. Section 5 considers the more general case where the coefficient of relative risk aversion differs from unity. Section 6 analyses the welfare gains from risk sharing and Section 7 briefly considers the implications of the model for the optimality of price targeting. Section 8 concludes the paper.

2 The Model

2.1 Market Structure

The world exists for a single period³ and consists of two countries, which will be referred to as the home country and the foreign country. Each country is populated by agents who consume a basket of goods containing all home and foreign produced goods.⁴ Each agent is a monopoly producer of a single differentiated product. There is a continuum of agents of unit mass in each country. Home agents are indexed $h \in [0, 1]$ and foreign agents are indexed $f \in [0, 1]$. All agents set prices in advance of the realisation of shocks and are contracted to meet demand at the pre-fixed prices.⁵ Prices are set in the currency of the producer.

The detailed structure of the home country is described below. The foreign country has an identical structure. Where appropriate, foreign real variables and foreign currency prices are indicated with an asterisk.

2.2 Preferences

All agents in the home economy have utility functions of the same form. The utility of agent z given by

$$U(z) = E\left[\frac{C(z)^{1-\rho}}{1-\rho} + \chi \log \frac{M(z)}{P} - Ky_i(z)\right]$$
(1)

³The model can easily be recast as a multi-period structure but this adds no significant insights. A true dynamic model, with multi-period nominal contracts and asset stock dynamics would be considerably more complex and would require much more extensive use of numerical methods. Newly developed numerical techniques are available to solve such models and this is likely to be an interesting line of future research (see Kim and Kim (2000), Sims (2000), Schmitt-Grohé and Uribe (2001) and Sutherland (2001)). However, the approach adopted in this paper yields useful insights which would not be available in a more complex model.

⁴In contrast to Obstfeld and Rogoff (2002) all goods in this model are traded goods. The presence of non-traded goods (or equivalently home bias in consumption preferences) is important in generating welfare gains from coordination in the Obstfeld and Rogoff model. The model presented in this paper generates gains to coordination when the elasticity of substitution between home and foreign goods differs from unity. These gains exist even when there are no non-traded goods.

⁵Obstfeld and Rogoff (2002) interpret their model as one where households supply labour to firms. They assume that each household is a monopoly supplier of a particular variety of labour and that wages are sticky (while goods prices are perfectly flexible). This is purely a matter of description. In terms of the analysis of this paper it makes no difference if households are described as supplying labour or supplying goods. In the first case it would be appropriate to regard wages as the sticky nominal variable, while in the second case it would be appropriate to regard prices as the sticky nominal variable.

where $\rho > 0$, C is a consumption index defined across all home and foreign goods, M denotes end-of-period nominal money holdings, P is the consumer price index, y(z) is the output of good z, E is the expectations operator, K is a log-normal stochastic labour-supply shock ($E[\log K] = 0$ and $Var[\log K] = \sigma_K^2$).

The consumption index C for home agents is defined as

$$C = \left[\left(\frac{1}{2}\right)^{\frac{1}{\theta}} C_H^{\frac{\theta-1}{\theta}} + \left(\frac{1}{2}\right)^{\frac{1}{\theta}} C_F^{\frac{\theta-1}{\theta}} \right]^{\frac{\theta}{\theta-1}}$$
(2)

where $\theta \ge 1$. C_H and C_F are indices of home and foreign produced goods defined as follows

$$C_{H} = \left[\int_{0}^{1} c_{H}(h)^{\frac{\phi-1}{\phi}} dh\right]^{\frac{\phi}{\phi-1}}, \quad C_{F} = \left[\int_{0}^{1} c_{F}(f)^{\frac{\phi-1}{\phi}} df\right]^{\frac{\phi}{\phi-1}}$$
(3)

where $\phi > 1$, $c_H(h)$ is consumption of home good h and $c_F(f)$ is consumption of foreign good f. The parameter θ is the elasticity of substitution between home and foreign goods. This is the key parameter which will be the focus of the analysis in later sections. In Obstfeld and Rogoff (2002) this parameter is fixed at unity.

The budget constraint of agent z is given by

$$M(z) = M_0 + (1+\alpha)p_H(z)y(z) - PC(z) - T$$
(4)

where M_0 and M(z) are initial and final money holdings, T is lump-sum government transfers, $p_H(z)$ is the price of home good z, α is a production subsidy and P is the aggregate consumer price index.

The government's budget constraint is

$$M - M_0 - \alpha P_H Y + T = 0 \tag{5}$$

where P_H is the aggregate price of home produced goods and Y is the aggregate output of the home economy, defined as follows

$$Y = C_H + C_H^* \tag{6}$$

where C_H^* is aggregate foreign demand for home goods.

2.3 Price Indices

The aggregate consumer price index for home agents is

$$P = \left[\frac{1}{2}P_{H}^{1-\theta} + \frac{1}{2}P_{F}^{1-\theta}\right]^{\frac{1}{1-\theta}}$$
(7)

where P_H and P_F are the price indices for home and foreign goods respectively defined as

$$P_{H} = \left[\int_{0}^{1} p_{H}(h)^{1-\phi} dh\right]^{\frac{1}{1-\phi}}, \quad P_{F} = \left[\int_{0}^{1} p_{F}(f)^{1-\phi} df\right]^{\frac{1}{1-\phi}}$$
(8)

The law of one price is assumed to hold. This implies $p_H(j) = p_H^*(j) S$ and $p_F(j) = p_F^*(j) S$ for all j where an asterisk indicates a price measured in foreign currency and S is the exchange rate (defined as the domestic price of foreign currency). Purchasing power parity holds in terms of aggregate consumer price indices, $P = P^*S$.

2.4 Consumption Choices

Individual home demand for representative home good, h, and foreign good, f, are given by

$$c_H(h) = C_H \left(\frac{p_H(h)}{P_H}\right)^{-\phi}, \quad c_F(f) = C_F \left(\frac{p_F(f)}{P_F}\right)^{-\phi} \tag{9}$$

where

$$C_H = \frac{1}{2}C\left(\frac{P_H}{P}\right)^{-\theta}, \quad C_F = \frac{1}{2}C\left(\frac{P_F}{P}\right)^{-\theta}$$
(10)

Foreign demands for home and foreign goods have an identical structure to the home demands. Individual foreign demand for representative home good, h, and foreign good, f, are given by

$$c_{H}^{*}(h) = C_{H}^{*}\left(\frac{p_{H}^{*}(h)}{P_{H}^{*}}\right)^{-\phi}, \quad c_{F}^{*}(f) = C_{F}^{*}\left(\frac{p_{F}^{*}(f)}{P_{F}^{*}}\right)^{-\phi}$$
(11)

where

$$C_{H}^{*} = \frac{1}{2} C^{*} \left(\frac{P_{H}^{*}}{P^{*}}\right)^{-\theta}, \quad C_{F}^{*} = \frac{1}{2} C^{*} \left(\frac{P_{F}^{*}}{P^{*}}\right)^{-\theta}$$
(12)

Each country has a population of unit mass so the total demands for goods are equivalent to individual demands.

2.5 Optimal Price Setting

Individual agents are each monopoly producers of a single differentiated good. They therefore set prices as a mark-up over marginal costs. The mark-up is given by $\phi/(\phi - 1)$. For convenience the mark-up is assumed to be offset by a production subsidy, α , which is paid to all producers (financed out of lump-sum taxes). The subsidy is set such that $\phi/[(\phi - 1)(1 + \alpha)] = 1$. This ensures that the expected level of output is at the socially optimal level (from the point of view of a world social planner).

The first-order condition for price setting is derived in the Appendix and implies the following

$$P_H = \frac{E\left[KY\right]}{E\left[Y/(PC^{\rho})\right]} \tag{13}$$

where Y is the total output of the home economy.

Notice that prices will contain a form of risk premium which will depend on the variances and covariances of the variables on the right hand side of (13). The risk premium reflects the fact that prices are set before shocks are realised. This risk premium plays a role in the link between shocks, monetary policy and welfare. An increase in the variance of KY for instance will (other things being equal) increase the risk premium and therefore increase the price of home produced goods. This lowers the expected level of output of home goods and therefore reduces the expected level of consumption for both home and foreign consumers. Home and foreign welfare is therefore reduced. Monetary policy can be used to affect the variances and covariances which determine the risk premium and can therefore also affect welfare.⁶

2.6 Home and Foreign Shocks

The foreign economy has a structure identical to the home economy. The foreign economy is subject to labour-supply shocks of the same form as the home economy. For simplicity it is assumed that the variances of the shocks are identical across the two countries, i.e.

$$\sigma_K^2 = \sigma_{K^*}^2 \tag{14}$$

The cross-country coefficient of correlation of shocks is given by υ where $-1 \leq \upsilon \leq 1.$

2.7 Money Demand and Supply

The first order condition for the choice of money holdings is

$$\frac{M}{P} = \chi C^{\rho} \tag{15}$$

It is assumed that the monetary authority in each country chooses a rule for the setting of the money supply. These rules may depend on the realisations of the supply shocks in each country and will take the form

$$M = M_0 K^{\delta_K} K^{*\delta_{K^*}} \text{ and } M^* = M_0^* K^{\delta_K^*} K^{*\delta_{K^*}^*}$$
(16)

⁶Note however that the risk premium is not the only link between monetary policy and welfare.

The feedback parameters δ_K , δ_{K^*} , δ_K^* and $\delta_{K^*}^*$ are chosen by policymakers before prices are set and shocks are realised. It is assumed that policymakers are able to pre-commit to their choice of rule.⁷

2.8 Financial Markets and Risk Sharing

When there are no financial markets the current account must balance in all states of the world, i.e.

$$P_H C_H^* = P_F C_F \tag{17}$$

where $P_H C_H^*$ is the value of home sales to the foreign country valued in home currency and $P_F C_F$ is the value of foreign sales to the home country valued in home currency.

In the risk sharing case it is assumed that sufficient contingent financial instruments exist to allow efficient sharing of consumption risks. This implies the following relationship

$$\frac{C^{\rho}}{C^{*\rho}} = \frac{P}{SP^*} \tag{18}$$

The fact that purchasing power parity also holds means that efficient risk sharing reduces to $C = C^*$.

It is important to specify the point in time at which agents are able to enter into risk-sharing contracts. There are two possible structures. In the first structure contingent claims markets open *after* policymakers have made their choice of monetary policy rules. In the second structure contingent claims markets open *before* policy rules have been chosen. The first structure implies a more limited form of insurance because agents can not insure against the choice of policy rules - they can only insure against the risk implied by a particular pair of rules. The analysis reported in the main text of this paper focuses on the first risk-sharing structure. The alternative structure is briefly analysed in the Appendix.

The distinction between the two risk-sharing structures is important from the point of view of policymakers. In the first structure policymakers are aware that agents are not fully insured against the potential negative impact of the choice of policy rule. Policymakers therefore internalise these costs. In the second case policymakers do not fully internalise the costs of policy rule choice. Not surprisingly

⁷In the case of coordinated policy it is not necessary to assume pre-commitment because the expected level of output is assumed to be at the socially optimal level from the point of view of a world social planner. But in the case of non-coordinated policymaking there is an incentive for individual country policymakers to attempt to reduce output *ex post* in order to improve the terms of trade. In the absence of pre-commitment this creates a deflationary bias in monetary policy. In this case no rational expectations equilibrium exists (as explained in Corsetti and Pesenti (2001b)).

this can greatly increase the cross-country spillover effects of monetary policymaking and can generate very large welfare gains from monetary policy coordination (as is shown in the Appendix). It is, however, questionable that the second risk-sharing structure is plausible. The choice of policy rules is not a stochastic event so it is not (strictly speaking) a source of risk. An insurance scheme which compensates consumers for the policy choices of their governments obviously creates a major moral hazard problem when governments act non-cooperatively.

3 Welfare

One of the main advantages of the model just described is that it provides a very natural and tractable measure of welfare which can be derived from the aggregate utility of agents. Following Obstfeld and Rogoff (1998, 2002) it is assumed that the utility of real balances is small enough to be neglected. It is therefore possible to measure aggregate welfare of home agents using the following

$$\Omega = E\left[\frac{C^{1-\rho}}{1-\rho} - KY\right]$$
(19)

It is not possible to derive an exact expression for welfare (except in special cases). The complication arising in this model (which does not arise in other models used in recent literature) is contained in equations (6) and (7). When θ is greater than unity neither of these equations is linear in logs. The model is therefore solved as a second-order approximation around a non-stochastic steady state. This allows a second-order accurate solution for welfare to be derived.

Define the non-stochastic steady state of the model to be the solution which results when $K = K^* = 1$ with $\sigma_K^2 = \sigma_{K^*}^2 = 0$ and for any variable X define $\hat{X} = \log(X/\bar{X})$ where \bar{X} is the value of variable X in the non-stochastic steady state.⁸ A second-order approximation of the welfare measure is given by

$$\tilde{\Omega} = E\left\{\hat{C} - \hat{Y} + \frac{1}{2}(1-\rho)\hat{C}^2 - \frac{1}{2}\left(\hat{Y} + \hat{K}\right)^2\right\} + O\left(\|\xi\|^3\right)$$
(20)

where $\tilde{\Omega}$ is the deviation in the level of welfare from the non-stochastic steady state and the term $O\left(\|\xi\|^3\right)$ contains all terms of third order and higher in deviations from the non-stochastic steady state. Notice that, to evaluate welfare, it is necessary to solve for both the first and second moments of output and consumption. The Appendix describes some of the details of the solution process.

It is now possible analyse the welfare gains from policy coordination.

⁸It is simple to show that the non-stochastic steady state will imply $\bar{Y} = \bar{C} = 1$.

4 The Welfare Gains from Policy Coordination: The Logarithmic Utility Case

It is useful first to consider the case where utility is logarithmic in consumption. In this case the coefficient of relative risk aversion, ρ , is set equal to unity.

4.1 Monetary Policy, the Exchange Rate and Output

Many of the implications of this model can be understood by examining the links between monetary policy in the two countries, the exchange rate and output. It is sufficient for this purpose to consider a log-linearised version of the model. First note that

$$\hat{P}_{H} = \hat{P}_{F}^{*} = 0 + O\left(\|\xi\|^{2}\right)$$
(21)

where $O\left(\|\xi\|^2\right)$ is a residual which contains all terms of order two and above. Equation (21) implies that the deviation of goods prices from their non-stochastic steady state values is zero (to a first-order approximation) so

$$\hat{P} = \frac{1}{2}\hat{S} + O\left(\|\xi\|^2\right), \quad \hat{P}^* = -\frac{1}{2}\hat{S} + O\left(\|\xi\|^2\right)$$
(22)

When these expressions are combined with the demands for home and foreign goods it is simple to show that home and foreign aggregate outputs are given by

$$\hat{Y} = \frac{1}{2} \left(\hat{C} + \hat{C}^* \right) + \frac{\theta}{2} \hat{S} + O\left(\|\xi\|^2 \right)$$
(23)

and

$$\hat{Y}^{*} = \frac{1}{2} \left(\hat{C} + \hat{C}^{*} \right) - \frac{\theta}{2} \hat{S} + O\left(\|\xi\|^{2} \right)$$
(24)

Thus aggregate output is determined by aggregate world consumption and the exchange rate. The exchange rate term is the expenditure switching effect. A depreciation of the exchange rate increases demand for home goods and reduces demand for foreign goods. Notice that the strength of the expenditure switching effect is determined by θ (which is the elasticity of substitution between home and foreign goods). These expressions hold regardless of the structure of financial markets.

Now consider the money market equations. When combined with the expressions for aggregate prices the money market equations imply

$$\hat{C} = \hat{M} + \frac{1}{2}\hat{S} + O\left(\|\xi\|^2\right), \quad \hat{C}^* = \hat{M}^* - \frac{1}{2}\hat{S} + O\left(\|\xi\|^2\right)$$
(25)

 \mathbf{SO}

$$\hat{C} + \hat{C}^* = \hat{M} + \hat{M}^* + O\left(\|\xi\|^2\right)$$
(26)

Thus aggregate world consumption is determined by the sum of home and foreign monetary policy. Again this expression holds regardless of the structure of financial markets.

The structure of financial markets comes into play in the determination of the exchange rate. When there is no financial market the current account has to balance in all states of the world. Using the expressions for aggregate prices and the demands for home and foreign goods, current account balance implies

$$\hat{S} = \frac{1}{\theta - 1} \left(\hat{C} - \hat{C}^* \right) + O\left(\|\xi\|^2 \right)$$
(27)

Thus, when home consumption exceeds foreign consumption the exchange rate must depreciate in order to maintain current account balance (and vice versa when foreign consumption exceeds home consumption). When this expression is combined with the expressions for aggregate consumption it is found that

$$\hat{S} = \frac{1}{\theta} \left(\hat{M} - \hat{M}^* \right) + O\left(\|\xi\|^2 \right)$$
(28)

Thus the exchange rate depends on relative money supplies.

When there is risk sharing the risk-sharing condition implies

$$\hat{S} = \hat{C} + \hat{P} - \hat{C}^* - \hat{P}^* \tag{29}$$

When combined with the money market relationships this implies

$$\hat{S} = \hat{M} - \hat{M}^* \tag{30}$$

Thus again the exchange rate depends on relative monetary supplies. But notice that the exchange rate is more sensitive to monetary policy when there is risk sharing (provided $\theta > 1$).

When the exchange rate expressions are combined with the expressions for aggregate consumption and outputs it is found that in the case of financial autarky

$$\hat{Y} = \hat{M} + O\left(\|\xi\|^2\right), \quad \hat{Y}^* = \hat{M}^* + O\left(\|\xi\|^2\right)$$
(31)

while in the case of risk sharing

$$\hat{Y} = \frac{1+\theta}{2}\hat{M} + \frac{1-\theta}{2}\hat{M}^* + O\left(\|\xi\|^2\right), \quad \hat{Y}^* = \frac{1+\theta}{2}\hat{M}^* + \frac{1-\theta}{2}\hat{M} + O\left(\|\xi\|^2\right)$$
(32)

The important point to note from these expressions is that in the financial autarky case monetary policy has no international spillover effects. A change in home monetary policy only affects home output and a change in foreign monetary policy only affects foreign output. This is because the effects of monetary policy on aggregate world demand are just enough to offset the expenditure switching effect. But in the risk sharing case monetary policy does have international spillover effects. In this case monetary policy has a larger effect on the exchange rate so the expenditure switching effect outweighs the effect of monetary policy on aggregate world consumption. Thus an increase in the home money supply causes an expansion of home output and a contraction of foreign output (and vice versa for an expansion of the foreign money supply).

The expressions for output and the exchange rate just derived will prove useful for understanding the source of the gains from coordination. The returns to monetary coordination are now analysed in the financial autarky and risk-sharing cases.

4.2 Financial Autarky

The Appendix shows that in this case home and foreign welfare can be written as follows

$$\tilde{\Omega} = -\frac{1}{4}E\left[\frac{1}{\theta}\left(2\theta - 1\right)\left(\hat{Y} + \hat{K}\right)^2 + \frac{1}{\theta}\left(\hat{Y}^* + \hat{K}^*\right)^2 + \frac{1}{2}\left(1 - \theta\right)\hat{S}^2\right]$$
(33)

and

$$\tilde{\Omega}^* = -\frac{1}{4}E\left[\frac{1}{\theta}\left(2\theta - 1\right)\left(\hat{Y}^* + \hat{K}^*\right)^2 + \frac{1}{\theta}\left(\hat{Y} + \hat{K}\right)^2 + \frac{1}{2}\left(1 - \theta\right)\hat{S}^2\right]$$
(34)

while the previous section showed that output levels and the exchange rate are linked to monetary policy by the following simple relationships

$$\hat{Y} = \hat{M}, \quad \hat{Y}^* = \hat{M}^*$$
 (35)

$$\hat{S} = \frac{1}{\theta} \left(\hat{M} - \hat{M}^* \right) \tag{36}$$

To simplify notation the residual terms $O\left(\|\xi\|^3\right)$ and $O\left(\|\xi\|^2\right)$ have been omitted from these and all subsequent expressions. It should be understood, however, that the welfare expressions are second-order approximations and the output and exchange rate expressions are first-order approximations.

The structure of the welfare functions can easily be understood. Notice that welfare depends negatively on the variances of $\hat{Y} + \hat{K}$ and $\hat{Y}^* + \hat{K}^*$. These terms are effectively the (log deviations of the) disutility of work effort for home and foreign producers. A higher variance of the disutility of work effort tends to raise the risk premium in goods prices. This reduces the expected level of output and consumption. Agents consume both home and foreign goods so welfare in both countries depends on the variance of the disutility of work effort in both countries. But notice that when $\theta > 1$ the variance of home disutility matters more for home welfare than does the variance of foreign disutility (and vice versa for foreign welfare). This is because a rise in the variance of home disutility not only raises the price of home goods for home agents it also results in a switch in world expenditure towards foreign goods and this reduces the income of home agents. The same mechanism means that the variance of foreign disutility has a greater impact on foreign welfare than the variance of home disutility.

Welfare depends positively on the variance of the exchange rate (when $\theta > 1$). This can be understood by considering the definition of the consumer price index. The consumer price index is concave in the price of home and foreign goods. Any volatility in the relative price of home and foreign goods (which would result from exchange rate volatility) will reduce the expected level of aggregate consumer prices. This has a positive effect on utility and welfare. (Another way to understand this effect is to note that, when home and foreign goods are substitutable, agents can reduce the average cost of their consumption basket by switching expenditure towards whichever set of goods are cheapest *ex post*. Relative price volatility is therefore a utility benefit.)

It is assumed that monetary authorities choose money supply rules of the following form

$$\hat{M} = \delta_K \hat{K} + \delta_{K^*} \hat{K}^* \tag{37}$$

and

$$\hat{M}^* = \delta_K^* \hat{K} + \delta_{K^*}^* \hat{K}^* \tag{38}$$

In the case of coordinated policymaking it is assumed that a single world monetary authority chooses the feedback parameters of both rules to maximise world welfare, where world welfare is given by the average of national welfare levels, i.e.

$$\tilde{\Omega}^W = \frac{1}{2} \left(\tilde{\Omega} + \tilde{\Omega}^* \right) \tag{39}$$

In the case of non-coordinated policymaking it is assumed that the feedback parameters of the home monetary rule are chosen by the home monetary authority in an attempt to maximise home welfare and the parameters of the foreign monetary rule are chosen by the foreign monetary authority in an attempt to maximise foreign welfare. Each monetary authority acts as a Nash player and takes as given the parameters of the other country's rule when choosing their own feedback parameters.

The coordinated equilibrium results in the following choices of feedback parameters

$$\delta_{K}^{C} = \delta_{K^{*}}^{*C} = \frac{-1 + \theta - 2\theta^{2}}{2\left(1 - \theta + \theta^{2}\right)}$$
(40)

$$\delta_{K^*}^C = \delta_K^{*C} = \frac{-1+\theta}{2\left(1-\theta+\theta^2\right)} \tag{41}$$

where the superscript C' indicates the coordinated equilibrium. The non-coordinated equilibrium results in

$$\delta_K^N = \delta_{K^*}^{*N} = \frac{1 - 3\theta + 4\theta^2}{-2\left(1 - 2\theta + 2\theta^2\right)} \tag{42}$$

$$\delta_{K^*}^N = \delta_K^{*N} = \frac{-1+\theta}{-2\left(1-2\theta+2\theta^2\right)}$$
(43)

where the superscript 'N' indicates the non-coordinated equilibrium. The world welfare level yielded by coordinated policy is

$$\tilde{\Omega}_A^C = \frac{(\theta - 1)}{4\left(1 - \theta + \theta^2\right)} (1 - \upsilon) \sigma_K^2 \tag{44}$$

where again the superscript 'C' indicates the coordinated equilibrium and the subscript 'A' indicates the financial autarky case. The welfare yielded by non-coordinated policy is

$$\tilde{\Omega}_{A}^{N} = \frac{\left(-2 + 7\theta - 9\theta^{2} + 4\theta^{3}\right)}{4\left(1 - 2\theta + 2\theta^{2}\right)^{2}}(1 - \upsilon)\sigma_{K}^{2}$$
(45)

As a point of reference it is useful to consider an inactive policy regime, where feedback parameters are all set to zero. (This is equivalent to a money targeting regime.) The welfare level yielded by this regime is

$$\tilde{\Omega}_A^M = -\frac{1}{2}\sigma_K^2 \tag{46}$$

where the superscript 'M' indicates the case of non-active policy (or money targeting). A number of propositions can now be established. (Proofs follow from a simple comparison of the above expressions and are omitted.)

Proposition 1 If v < 1 and $\theta > 1$ then $\tilde{\Omega}_A^C > \tilde{\Omega}_A^N$, i.e. there are gains from coordination.

It is clear from expressions (33) to (36) that there will be gains to coordination provided $\theta > 1$. When $\theta > 1$ each monetary authority cares about the variance of the exchange rate, and monetary policy in each country affects the exchange rate. In addition, when $\theta > 1$, each monetary authority cares more about the volatility of the disutility of work effort in its own country than it does about the volatility of the disutility of work effort in the other country. There is therefore a policy spillover (operating through the exchange rate) and an incentive to bias policy to the benefit of domestic welfare. The gains from coordination disappear in two circumstances. The first case is when $\theta = 1$. In this case exchange rate volatility does not affect welfare so there is no policy spillover. Each monetary authority therefore maximises the welfare of its population by minimising the variance of the disutility of work effort in its own country. This also maximises world welfare. The second case where there are no gains from coordination is when the shocks in the two countries are perfectly correlated, i.e. when v = 1. This corresponds to a result noted and emphasised by Obstfeld and Rogoff (2002). When shocks are perfectly correlated the use of monetary policy to stabilise the disutility of work effort in one country will automatically also stabilise disutility of work effort in the other country. There is therefore no difference between coordinated and non-coordinated policymaking.

Proposition 2 If $\theta > 1$ then: (a) $|\delta_K^C| = |\delta_{K^*}^{*C}| > |\delta_K^N| = |\delta_{K^*}^{*N}|$ and $|\delta_{K^*}^C| = |\delta_{K^*}^{*C}| > |\delta_{K^*}^{N}| = |\delta_{K^*}^{*N}|$ and (b) $Var\left(\hat{S}^N\right) < Var\left(\hat{S}^C\right)$, $Var\left(\hat{Y}^N\right) < Var\left(\hat{Y}^C\right)$ and $Var\left(\hat{Y}^{*N}\right) < Var\left(\hat{Y}^{*C}\right)$ (where the superscripts 'C' and 'N' indicate values in coordinated and non-coordinated equilibria respectively).

This proposition shows that non-coordinated policymaking is less active than coordinated policymaking. It also shows that the exchange rate and output levels are less volatile with non-coordinated policymaking. In other words non-coordinated policymaking has a bias towards over-stabilisation. At first sight it may seem strange that optimal coordinated policy should produce more volatility of output and the exchange rate (and by implication the terms of trade). But it should be born in mind that labour supply shocks imply that the socially optimal level of output is changing. The socially optimal monetary policy should allow these changes to occur. Non-coordinated policymaking is preventing full adjustment of real quantities to the underlying socially optimal levels.⁹

Proposition 1 establishes that there are gains to coordination when $\theta > 1$. But in order to determine the size of these gains it is necessary to perform some numerical exercises with different values of θ . Table 1 reports some values for welfare with $\sigma_K^2 = \sigma_{K^*}^2 = 0.01$ and v = 0. A range of values of θ has been suggested in previous literature, for instance Benigno and Benigno (2001a) suggest $\theta = 6$. Table 1 shows welfare calculations for $\theta = 1$ to $\theta = 10$. The first row shows the welfare gain from coordinated policy relative to an inactive policy (i.e. $\tilde{\Omega}_A^C - \tilde{\Omega}_A^M$). The figures in the first row therefore represent the maximum possible gain from following an active policy. The second row shows the welfare gain from non-coordinated policy relative

⁹This result is discussed in more detail in Sutherland (2002).

θ	1	2	4	6	8	10
$\tilde{\Omega}_A^C - \tilde{\Omega}_A^M$	0.500	0.583	0.558	0.540	0.531	0.525
$\tilde{\Omega}^N_A{-}\tilde{\Omega}^M_A$	0.500	0.580	0.555	0.539	0.530	0.524
$ ilde{\Omega}^{C}_{A} {-} ilde{\Omega}^{N}_{A}$		0.003	0.002	0.001	0.001	0.001
$\frac{100{\times}\left(\tilde{\Omega}_{A}^{C}{-}\tilde{\Omega}_{A}^{N}\right)}{\left(\tilde{\Omega}_{A}^{C}{-}\tilde{\Omega}_{A}^{M}\right)}$	0.0	0.6	0.4	0.2	0.2	0.1

Table 1: The welfare effects of coordination: Financial autarky

to an inactive policy (i.e. $\tilde{\Omega}_A^N - \tilde{\Omega}_A^M$). The third row shows the absolute gains from coordination (i.e. $\tilde{\Omega}_A^C - \tilde{\Omega}_A^N$). In each case these figures are measured as a percentage of steady state consumption. The fourth row shows the gains from coordination as a percentage of the maximum possible gain from an active policy (i.e. row 3 as a percentage of row 1). It is apparent from Table 1 that the welfare gain from coordination is positive when θ is greater than unity. But the gain is never large, either in absolute or relative terms. This is very similar to the result emphasised by Obstfeld and Rogoff (2002).

4.3 Risk Sharing

The procedure described in the Appendix can be used to show that home and foreign welfare in the risk-sharing case can be written as follows

$$\tilde{\Omega} = -\frac{1}{4}E\left[\frac{1}{\theta}\left(2\theta - 1\right)\left(\hat{Y} + \hat{K}\right)^2 + \frac{1}{\theta}\left(\hat{Y}^* + \hat{K}^*\right)^2 + \frac{1}{2}\theta\left(1 - \theta\right)\hat{S}^2\right]$$
(47)

and

$$\tilde{\Omega}^* = -\frac{1}{4}E\left[\frac{1}{\theta}\left(2\theta - 1\right)\left(\hat{Y}^* + \hat{K}^*\right)^2 + \frac{1}{\theta}\left(\hat{Y} + \hat{K}\right)^2 + \frac{1}{2}\theta\left(1 - \theta\right)\hat{S}^2\right]$$
(48)

while it was shown above that output levels and the exchange rate are linked to monetary policy by the following simple relationships

$$\hat{Y} = \frac{1+\theta}{2}\hat{M} + \frac{1-\theta}{2}\hat{M}^*, \quad \hat{Y}^* = \frac{1+\theta}{2}\hat{M}^* + \frac{1-\theta}{2}\hat{M}$$
(49)

$$\hat{S} = \hat{M} - \hat{M}^* \tag{50}$$

The form of the welfare function for each country is almost identical to the autarky case. The only difference is a small change to the coefficient on the variance of the exchange rate. The main difference between the this case and the previous case is contained in the determination of output. There is now a spillover effect from monetary policy in one country to the level of output in the other country. It is clear that this creates more scope for gains from coordinated policy. The quantitative implications of this spillover effect are considered below. First consider the expressions for feedback coefficients and welfare levels.

The coordinated equilibrium results in the following choices of feedback parameters

$$\delta_K^C = \delta_{K^*}^{*C} = -1 \tag{51}$$

$$\delta_{K^*}^C = \delta_K^{*C} = 0 \tag{52}$$

while the non-coordinated equilibrium results in

$$\delta_K^N = \delta_{K^*}^{*N} = \frac{1 - 3\theta^2}{-2\theta \left(1 - 2\theta\right)}$$
(53)

$$\delta_{K^*}^N = \delta_K^{*N} = \frac{1 - 2\theta + \theta^2}{2\theta \left(1 - 2\theta\right)} \tag{54}$$

The world welfare level yielded by coordinated policy is

$$\tilde{\Omega}_R^C = \frac{(\theta - 1)}{4} (1 - \upsilon) \sigma_K^2 \tag{55}$$

where the subscript 'R' indicates the risk sharing case. The welfare yielded by noncoordinated policy is

$$\tilde{\Omega}_R^N = \frac{\left(-1 + 3\theta - \theta^2 - 4\theta^3 + 3\theta^4\right)}{4\theta \left(1 - 2\theta\right)^2} (1 - \upsilon)\sigma_K^2 \tag{56}$$

Again, as a point of reference it is useful to consider an inactive policy regime. The welfare level yielded by this regime is

$$\tilde{\Omega}_R^M = -\frac{1}{2}\sigma_K^2 \tag{57}$$

A number of propositions can now be established (and again the proofs are omitted).

Proposition 3 If v < 1 and $\theta > 1$ then $\tilde{\Omega}_R^C > \tilde{\Omega}_R^N$, i.e. there are gains from coordination.

It is clear from the expressions (47) to (50) that gains from coordination will arise. All the factors that were present in the autarky case are also present in this case. When $\theta > 1$ both monetary authorities care about the volatility of the exchange rate and both monetary authorities can affect the exchange rate using monetary policy. Also welfare in each country is affected more by the volatility of the disutility of work effort within the country than the volatility in the other country. But now there is an extra spillover effect of monetary policy. When $\theta > 1$ a monetary expansion in the home country reduces output in the foreign country because of the expenditure switching effect. Likewise a monetary expansion in the foreign country reduces output in the home country.

Again notice that there are two cases where the gains from coordination disappear. The first is where $\theta = 1$. In this case the spillover effect from monetary policy to foreign output disappears. The second case is where v = 1. Correlated shocks do not create any conflicts between optimal policy in each country. (Note, however, this is not true when the degree of risk aversion is different from unity. This case will be considered in the next section.)

Proposition 4 If $\theta > 1$ then: (a) $|\delta_K^C| = |\delta_{K^*}^{*C}| > |\delta_K^N| = |\delta_{K^*}^{*N}|$ and $|\delta_{K^*}^N| = |\delta_K^{*N}| > |\delta_K^{C*}| = |\delta_K^{*C}|$ and (b) $Var\left(\hat{S}^N\right) < Var\left(\hat{S}^C\right)$, $Var\left(\hat{Y}^N\right) < Var\left(\hat{Y}^C\right)$ and $Var\left(\hat{Y}^{*N}\right) < Var\left(\hat{Y}^{*C}\right)$.

In the autarky case it was clear that non-coordinated policy was less active than coordinated policymaking. In this case coordinated policymaking implies a stronger monetary policy reaction to shocks occurring within a country but a smaller reaction to shocks occurring in the other country. In other words non-coordinated policy involves a shifting of the burden of policy adjustment onto the other country. It remains true however that non-coordinated policy implies less volatility in the exchange rate and output levels.

The quantitative implications of risk sharing for the gains from coordination are illustrated in Table 2. The parameter values are the same as those used to construct Table 1 and the structure of the table is identical. It is apparent that the gains from coordination are much larger than in the autarky case in both absolute and relative terms. For instance when $\theta = 6$ the gains from coordination are worth 0.2 percent of steady state consumption which represents 12.3 percent of the gains from optimal stabilisation. These figures obviously can not be described as large, but they are also not trivial.¹⁰

¹⁰Notice from (55), (56) and (57) that the size of the welfare effects is proportional to the aggregate variance of the shocks. In a more general model, with more sources of shocks and some persistence in the shock processes, the size of the welfare effects will depend on some aggregate of all shock variances and the degree of persistence of the shocks. This may generate larger welfare effects than reported here.

θ	1	2	4	6	8	10
$ ilde{\Omega}_R^C - ilde{\Omega}_R^M$						
$ ilde{\Omega}_R^N {-} ilde{\Omega}_R^M$	0.500	0.736	1.147	1.535	1.916	2.296
$ ilde{\Omega}_R^C {-} ilde{\Omega}_R^N$		0.014	0.103	0.215	0.334	0.454
$\frac{100{\times}\left(\tilde{\Omega}_{R}^{C}{-}\tilde{\Omega}_{R}^{N}\right)}{\left(\tilde{\Omega}_{R}^{C}{-}\tilde{\Omega}_{R}^{M}\right)}$	0.0	1.8	8.3	12.3	14.8	16.5

Table 2: The welfare effects of coordination: Risk sharing

5 Risk Aversion and the Welfare Gains from Policy Coordination

The analysis so far has focused on the case where utility is logarithmic in consumption. The coefficient of relative risk aversion, ρ , is therefore unity. This section considers the implications of varying the degree of risk aversion. In what follows much of the explicit derivation is omitted and the discussion concentrates on the parts of the analysis which are modified by the allowing for $\rho \neq 1$.

5.1 Financial Autarky

Using the procedure described in the Appendix it is possible to show that home and foreign welfare can now be written as follows

$$\tilde{\Omega} = -E \left[\frac{\omega \left(\hat{Y} + \hat{K} \right)^2 + \left(\hat{Y}^* + \hat{K}^* \right)^2 + \left(\hat{Y} - \hat{M} \right)^2 - \left(\hat{Y}^* - \hat{M}^* \right)^2}{4 \left[1 + \rho \left(\theta - 1 \right) \right]} + \frac{1}{8} \left(1 - \theta \right) \hat{S}^2 - \frac{1}{2} \left(1 - \rho \right) \hat{C}^2 \right]$$
(58)

and

$$\tilde{\Omega}^{*} = -E \left[\frac{\omega \left(\hat{Y}^{*} + \hat{K}^{*} \right)^{2} + \left(\hat{Y} + \hat{K} \right)^{2} + \left(\hat{Y}^{*} - \hat{M}^{*} \right)^{2} - \left(\hat{Y} - \hat{M} \right)^{2}}{4 \left[1 + \rho \left(\theta - 1 \right) \right]} + \frac{1}{8} \left(1 - \theta \right) \hat{S}^{2} - \frac{1}{2} \left(1 - \rho \right) \hat{C}^{*2} \right]$$
(59)

where $\omega = [1 + 2\rho (\theta - 1)]$. The output levels and the exchange rate are linked to monetary policy by the following simple relationships

$$\hat{Y} = \frac{\left[1 + \rho \left(2\theta - 1\right)\right] \hat{M} + \left(1 - \rho\right) \hat{M}^*}{2\rho \left[1 + \rho \left(\theta - 1\right)\right]}, \quad \hat{Y}^* = \frac{\left[1 + \rho \left(2\theta - 1\right)\right] \hat{M}^* + \left(1 - \rho\right) \hat{M}}{2\rho \left[1 + \rho \left(\theta - 1\right)\right]} \quad (60)$$

$$\hat{S} = \frac{1}{1 + \rho \left(\theta - 1\right)} \left(\hat{M} - \hat{M}^*\right)$$
(61)

It is also necessary to consider the relationship between monetary policy and consumption levels. These are as follows

$$\hat{C} = \frac{\left[1 + 2\rho\left(\theta - 1\right)\right]\hat{M} + \hat{M}^{*}}{2\rho\left[1 + \rho\left(\theta - 1\right)\right]}, \quad \hat{C}^{*} = \frac{\left[1 + 2\rho\left(\theta - 1\right)\right]\hat{M}^{*} + \hat{M}}{2\rho\left[1 + \rho\left(\theta - 1\right)\right]} \tag{62}$$

It is apparent from (58) and (59) that the relationship between welfare and output and exchange rate volatility is rather more complicated than in the $\rho = 1$ case. Equations (60) and (61) show that the relationship between output and the exchange rate and monetary policy is also more complicated than in the $\rho = 1$ case. It is now apparent from (60) and (61) that monetary policy can have a spillover effect on output even when there is no risk sharing. Thus, when $\rho > 1$ an expansion of the home money supply will have a contractionary effect on foreign output (and vice versa for an expansion of the foreign money supply). This spillover effect creates a new possibility for welfare gains from policy coordination.

Another important new feature of the above relationships is that welfare now depends on the volatility of consumption. The reason for this is obvious. When utility is logarithmic in consumption agents do not care about the variance of the log-deviation of consumption. But when $\rho > 1$ risk aversion is sufficiently strong to imply that volatility of the log-deviation of consumption has a negative effect on aggregate utility (and vice versa for $\rho < 1$).

Equations (62) show how consumption depends on monetary policy. It is apparent that when $\rho \neq 1$ and $\theta > 1$ consumption in each country depends on monetary policy in each country. Thus, for instance, an increase in the home money supply raises both home and foreign consumption. This is because an increase in the home money supply raises output and income of home agents and this allows home agents to increase consumption of both home and foreign goods. This raises the income of foreign agents who are thus able also to raise consumption. Notice however that the increase in the home money supply has a larger effect on home consumption than it does on foreign consumption.

This link between the money supply in one country and the level of consumption in the other country creates another a new spillover effect of monetary policy. This again creates potential welfare gains from monetary policy coordination.

Table 3 illustrates the quantitative implications of these new spillover effects. In this table $\theta = 2$ and the value of ρ is varied between 1/4 and 8. The baseline parameter values are the same as in previous examples. It is apparent that the size of the welfare gain is increasing as the degree of risk aversion deviates from unity. The size of the welfare gain is now rather larger in relative terms but it remains small in

ρ	1/4	1/2	1	2	4	6	8
$ ilde{\Omega}^C_A - ilde{\Omega}^M_A$	1.444	0.900	0.583	0.375	0.229	0.167	0.131
$\tilde{\Omega}^N_A{-}\tilde{\Omega}^M_A$	1.421	0.898	0.580	0.359	0.208	0.147	0.113
$ ilde{\Omega}^{C}_{A}{-} ilde{\Omega}^{N}_{A}$	0.023	0.002	0.003	0.016	0.021	0.020	0.018
$\frac{\frac{100 \times \left(\tilde{\Omega}_{A}^{C} - \tilde{\Omega}_{A}^{N}\right)}{\left(\tilde{\Omega}_{A}^{C} - \tilde{\Omega}_{A}^{N}\right)}}$	1.589	0.227	0.571	4.167	9.276	12.00	13.64

Table 3: The welfare effects of coordination: Financial autarky

absolute terms. Thus the new spillover effects working via output and consumption are limited in magnitude when there is no risk sharing.

5.2 Risk sharing

Using the solution procedure described in the Appendix it is possible to show that home and foreign welfare in the risk-sharing case can be written as follows

$$\tilde{\Omega} = -E \left[\frac{\omega \left(\hat{Y} + \hat{K} \right)^2 + \left(\hat{Y}^* + \hat{K}^* \right)^2 + \left(\hat{Y} - \hat{M} \right)^2 - \left(\hat{Y}^* - \hat{M}^* \right)^2}{4 \left[1 + \rho \left(\theta - 1 \right) \right]} + \frac{1}{8} \left(1 - \theta \right) \theta \hat{S}^2 - \frac{1}{2} \left(1 - \rho \right) \hat{C}^2 \right]$$
(63)

and

$$\tilde{\Omega}^{*} = -E \left[\frac{\omega \left(\hat{Y}^{*} + \hat{K}^{*} \right)^{2} + \left(\hat{Y} + \hat{K} \right)^{2} + \left(\hat{Y}^{*} - \hat{M}^{*} \right)^{2} - \left(\hat{Y} - \hat{M} \right)^{2}}{4 \left[1 + \rho \left(\theta - 1 \right) \right]} + \frac{1}{8} \left(1 - \theta \right) \theta \hat{S}^{2} - \frac{1}{2} \left(1 - \rho \right) \hat{C}^{*2} \right]$$
(64)

where again $\omega = [1 + 2\rho (\theta - 1)]$. Output levels, the exchange rate and consumption are linked to monetary policy by the following simple relationships

$$\hat{Y} = \frac{1+\rho\theta}{2\rho}\hat{M} + \frac{1-\rho\theta}{2\rho}\hat{M}^*, \quad \hat{Y}^* = \frac{1+\rho\theta}{2\rho}\hat{M}^* + \frac{1-\rho\theta}{2\rho}\hat{M}$$
(65)

$$\hat{S} = \hat{M} - \hat{M}^* \tag{66}$$

$$\hat{C} = \frac{\hat{M} + \hat{M}^*}{2\rho}, \quad \hat{C}^* = \frac{\hat{M} + \hat{M}^*}{2\rho}$$
(67)

It is again apparent that the main difference between these relationships and their counterparts in the $\rho = 1$ case is the fact that welfare depends on the variance of consumption. This again creates a new spillover effect. But notice now that consumption in each country depends equally on both home and foreign monetary policy. This is an obvious consequence of risk sharing. It is therefore possible that

ρ	/	/		2			8
$ ilde{\Omega}_R^C {-} ilde{\Omega}_R^M$							
$\tilde{\Omega}_R^N{-}\tilde{\Omega}_R^M$							
$ ilde{\Omega}_R^C {-} ilde{\Omega}_R^N$		0.111	0.014	0.089	0.456	0.898	1.366
$\frac{100 \times \left(\tilde{\Omega}_{R}^{C} - \tilde{\Omega}_{R}^{N}\right)}{\left(\tilde{\Omega}_{R}^{C} - \tilde{\Omega}_{R}^{M}\right)}$	22.98	11.11	1.852	14.24	81.04	165.8	257.1

Table 4: The welfare effects of coordination: Risk sharing

ρ	1/4	1/2	1	2	4	6	8
$ ilde{\Omega}_R^C - ilde{\Omega}_R^M$							
$\tilde{\Omega}_R^N{-}\tilde{\Omega}_R^M$	0.798	0.444	0.250	0.038	-0.393	-0.857	-1.335
$\tilde{\Omega}_{R}^{C} - \tilde{\Omega}_{R}^{N}$	0.203	0.056	0.000	0.087	0.456	0.898	1.366
$\tfrac{100 \times \left(\tilde{\Omega}_R^C - \tilde{\Omega}_R^N\right)}{\left(\tilde{\Omega}_R^C - \tilde{\Omega}_R^M\right)}$					729.0		4371

Table 5: The welfare effects of coordination: Symmetric shocks

the spillover effect operating through consumption levels is potentially more significant than in the autarky case (where home monetary policy had a greater effect on home consumption and foreign monetary policy had a greater effect on foreign consumption).

Table 4 illustrates the quantitative implications for the gains to coordination. It is immediately apparent that the gains from coordination can now be quite large, both in absolute and relative terms, even for quite moderate values of θ and ρ . For instance when $\rho = 4$ the gains from coordination are approaching 0.5 percent of steady state consumption which represents over 80 percent of the gains from optimal stabilisation policy.

5.3 Symmetric Shocks and the Gains from Policy Coordination

In Section 4 is was shown that with $\rho = 1$ there were no welfare gains to policy coordination when the shocks in the two countries are perfectly correlated. This continues to be true in the autarky case when $\rho \neq 1$. But it is not true on the risk-sharing case when $\rho \neq 1$. Table 5 reports the welfare figures for the risk-sharing case when shocks are perfectly correlated (i.e. v = 1).¹¹ It is apparent that there are substantial gains from coordination. This contrasts with a result emphasised by Obstfeld and Rogoff (2002).

¹¹In this example the variances of the individual country shocks are $\sigma_K^2 = \sigma_{K^*}^2 = 0.005$. This ensures that the aggregate world variance is identical to the previous examples where v = 0.

6 The Welfare Gains from Risk Sharing

The analysis so far has concentrated on the welfare gains from policy coordination. But the model also yields estimates of the welfare gains from risk sharing. Table 6 repeats some of numerical welfare results from the previous sections in a way which allows a comparison across financial market structures. Table 6 focuses on the effects of varying θ when $\rho = 1$ (i.e. the case of logarithmic utility). The first row shows the welfare gains from risk sharing when monetary policy is coordinated. The second row shows the same results for the non-coordinated policy regime.

It is clear from expressions (44) and (55) that there is an unambiguous welfare gain to risk sharing when policy is coordinated (provided $\theta > 1$ and v < 1). That there should be such a welfare gain is not *a priori* obvious in a model where there are several market distortions (such as monopoly power and sticky nominal prices). The figures in the first row in Table 6 provide a quantitative measure of the potential welfare gain from risk sharing. These figures are within the range of estimates suggested by previous literature.¹²

The welfare effects of risk sharing are somewhat smaller when monetary policy is not coordinated. A comparison of (45) and (56) shows that risk sharing again provides an unambiguous welfare gain when $\rho = 1$. But figures in the second row of Table 6 show that the welfare gain is smaller than when monetary policy is coordinated. The monetary policy spillover effects created by risk sharing partly offset the welfare gains of risk sharing when monetary policy is uncoordinated.

The welfare gains from risk sharing are, however, very sensitive to the degree of risk aversion. This is illustrated in Table 7. This table again reports values for the welfare gain from risk sharing for the cases of coordinated and non-coordinated monetary policy. In this case θ is set equal to 2 and ρ is varied. It is again clear that the welfare effect of risk sharing is positive when monetary policy is coordinated. But it is also now apparent that the welfare effect can be negative when monetary

 $^{^{12}}$ For instance Cole and Obstfeld (1991) suggest a welfare gain from risk sharing of the order of 0.2 percent of steady state consumption while van Wincoop (1994) suggests a gain closer to 5 percent of steady state consumption.

θ	1	2	4	6	8	10
$\tilde{\Omega}_{R}^{C}-\tilde{\Omega}_{A}^{C}$	0.000	0.167	0.692	1.210	1.719	2.225
$\tilde{\Omega}_R^N{-}\tilde{\Omega}_A^N$	0.000	0.156	0.592	0.996	1.387	1.772

Table 6: The welfare effects of risk sharing: Logarithmic utility

ρ	1/4	1/2	1	2	4	6	8
$\tilde{\Omega}_{R}^{C}-\tilde{\Omega}_{A}^{C}$	0.056	0.100	0.167	0.250	0.333	0.375	0.400
$\tilde{\Omega}_R^N - \tilde{\Omega}_A^N$	-0.266	-0.009	0.156	0.177	-0.101	-0.503	-0.948

Table 7: The welfare effects of risk sharing: Risk aversion

policy is uncoordinated. Thus the monetary policy spillovers created by risk sharing can be so strong that they outweigh the welfare benefits of risk sharing. The figures in Table 7 suggest that this can occur for quite moderate values of ρ and θ .

7 The Optimality of Price Targeting

One theme in the recent literature on monetary policy has been the welfare implications of price (or inflation) targeting. A number of authors have argued that price or inflation targeting is desirable from a welfare point of view (see for instance, King and Wolman (1999), Goodfriend and King (2001), Woodford (2001)). The final section of this paper briefly discusses the implications of the model for the optimality of price targeting.

The model assumes that all prices are fixed in advance so it is not possible directly to analyse a price targeting policy. But it is possible to gain some indirect insight into the implications for prices by considering the first-order condition for price setting that would be relevant if agents were able to set prices after shocks are realised. The first-order condition for the choice of prices in a flexible-price equilibrium is derived in the Appendix and implies the following

$$\hat{P}_H = \hat{K} + \hat{P} + \rho \hat{C}, \quad \hat{P}_F^* = \hat{K}^* + \hat{P}^* + \rho \hat{C}^*$$
(68)

A price targeting policy implies $\hat{P}_H = \hat{P}_F^* = 0$ so, when expressions (68) are combined with the money demand equations the following monetary rules are obtained

$$\hat{M} = -\hat{K}, \quad \hat{M}^* = -\hat{K}^*$$
(69)

These rules are relevant for all values of θ and ρ and for all financial market structures. So any equilibrium which implies policy rules of the above form is consistent with price targeting.

It is immediately clear that neither coordinated nor non-coordinated policymaking is consistent with price targeting in the case of financial autarky (as is argued in Benigno (2001) and Obstfeld and Rogoff (2002)). It is however clear that coordinated policy is consistent with price targeting when there is risk sharing. Uncoordinated policymaking is only consistent with price targeting in the risk-sharing case for particular parameter combinations (as shown in Benigno and Benigno (2001a).

8 Conclusion

This paper has analysed the welfare effects of monetary policy coordination in a model where the elasticity of substitution between home and foreign goods can differ from unity. It is shown that welfare gains to policy coordination can arise when the elasticity is greater than unity, but these gains are quantitatively small when there is no international financial market. When, however, there is a sufficiently sophisticated financial market to allow full consumption risk sharing the gains from policy coordination are found to be much larger. This is particularly true when the coefficient of relative risk aversion differs from unity.

The model also yields results concerning the welfare impact of financial market integration (i.e. a move from financial autarky to risk sharing). It is found that the additional monetary policy spillover effects created by financial markets can be so strong that financial market integration can have a negative impact on welfare if monetary policy is not coordinated.

This paper has considered two extreme forms of financial market structure. The gains from coordination are found to differ significantly between the two extremes. But neither extreme is entirely satisfactory as a representation of reality. An obvious next step in this line of research is to investigate the welfare gains to coordination in some intermediate financial market structure. A possible example of an intermediate structure is one where financial trade only takes place in the form of non-contingent bonds. This type of model will inevitably involve asset stock dynamics and will therefore require more extensive use of numerical simulation techniques.¹³ An alternative way to model an intermediate degree of risk sharing has recently been proposed by Ligon, Thomas and Worrall (1997, 2000) and Kehoe and Perri (2000). It may also be interesting to consider the gains from monetary policy coordination in this alternative 'endogenous incomplete market' framework.

¹³Techniques which make this form of analysis possible have recently been developed by Kim and Kim (2000), Sims (2000), Schmitt-Grohé and Uribe (2001) and Sutherland (2001).

Appendix

Optimal Price Setting

The price-setting problem facing a fixed-price producer is the following:

$$MaxU(z) = E\left\{\frac{C^{1-\rho}(z)}{1-\rho} + \log\frac{M}{P} - Ky(z)\right\}$$
(70)

subject to

$$PC(z) = (1+\alpha) p_H(z) y(z) + M_0 - M - T$$
(71)

$$y(z) = c_H(z) + c_H^*(z) = (C_H + C_H^*) \left(\frac{p_H(z)}{P_H}\right)^{-\phi}$$
(72)

The first order condition with respect to $p_{H}(z)$ is

$$E\left\{(1+\alpha)\frac{y(z)}{PC^{\rho}(z)} - \phi\left[(1+\alpha)\frac{p_{H}(z)}{PC^{\rho}(z)} - K\right]\frac{y(z)}{p_{H}(z)} = 0\right\} = 0$$
(73)

In equilibrium all agents choose the same price and consumption level so

$$E\left\{(1+\alpha)\frac{Y}{PC^{\rho}} - \phi\left[(1+\alpha)\frac{P_H}{PC^{\rho}} - K\right]\frac{Y}{P_H} = 0\right\} = 0$$
(74)

where

$$Y = C_H + C_H^* \tag{75}$$

Rearranging yields the expression in the main text.

The price-setting problem facing a flexible-price producer is the following:

$$MaxU(z) = \frac{C^{1-\rho}(z)}{1-\rho} + \log\frac{M}{P} - Ky(z)$$
(76)

subject to

$$PC(z) = (1 + \alpha) p_H(z) y(z) + M_0 - M - T$$
(77)

$$y(z) = c_H(z) + c_H^*(z) = (C_H + C_H^*) \left(\frac{p_H(z)}{P_H}\right)^{-\phi}$$
(78)

The first order condition with respect to $p_{H}(z)$ is

$$(1+\alpha)\frac{y(z)}{PC^{\rho}(z)} - \phi \left[(1+\alpha)\frac{p_{H}(z)}{PC^{\rho}(z)} - K \right] \frac{y(z)}{p_{H}(z)} = 0$$
(79)

In equilibrium all agents choose the same price and consumption level so

$$(1+\alpha)\frac{Y}{PC^{\rho}} - \phi\left[(1+\alpha)\frac{P_H}{PC^{\rho}} - K\right]\frac{Y}{P_H} = 0$$
(80)

where

$$Y = C_H + C_H^* \tag{81}$$

Rearranging yields the expression in the main text.

Model Solution

The solution procedure is described using the autarky case as an illustration. The amendments necessary to derive the risk sharing solution are then described.

In order to derive a solution for the welfare measure it is necessary to derive solutions for both the first and second moments of the model. The first step in the solution process is to replace each equation of the model with a second-order approximation in terms of log-deviations from the non-stochastic steady state. Most of the equations of the model are linear in logs so this process does not involve any approximation for those equations. There are just three pairs of equations where approximations are necessary.

The log-deviation form of the money market equations implies

$$\hat{M} = \hat{P} + \rho \hat{C}, \quad \hat{M}^* = \hat{P}^* + \rho \hat{C}^*$$
(82)

For home and foreign demand equations the log-deviation forms are

$$\hat{C}_H = \hat{C} - \theta \left(\hat{P}_H - \hat{P} \right), \quad \hat{C}_F = \hat{C} - \theta \left(\hat{P}_F - \hat{P} \right)$$
(83)

and

$$\hat{C}_{H}^{*} = \hat{C}^{*} - \theta \left(\hat{P}_{H}^{*} - \hat{P}^{*} \right), \quad \hat{C}_{F}^{*} = \hat{C}^{*} - \theta \left(\hat{P}_{F}^{*} - \hat{P}^{*} \right)$$
(84)

The log-deviation form of current account balance implies

$$\hat{P}_{H} + \hat{C}_{H}^{*} = \hat{P}_{F} + \hat{C}_{F} \tag{85}$$

And the log-deviation form of purchasing power parity implies

$$\hat{P} = \hat{P}^* + \hat{S} \tag{86}$$

None of the above equations require any approximation when converting to logdeviation form.

The expressions for total outputs, aggregate prices and price setting do require approximation. The second-order approximation for the total output equations are

$$\hat{Y} = \frac{1}{2}\hat{C}_{H} + \frac{1}{2}\hat{C}_{H}^{*} + \lambda_{Y} + O\left(\|\xi\|^{3}\right), \quad \hat{Y}^{*} = \frac{1}{2}\hat{C}_{F} + \frac{1}{2}\hat{C}_{F}^{*} + \lambda_{Y^{*}} + O\left(\|\xi\|^{3}\right) \quad (87)$$

where

$$\lambda_Y = \frac{1}{8} \left(\hat{C}_H - \hat{C}_H^* \right)^2, \quad \lambda_{Y^*} = \frac{1}{8} \left(\hat{C}_F - \hat{C}_F^* \right)^2$$

The second-order approximations for the aggregate price indices are

$$\hat{P} = \frac{1}{2}\hat{P}_{H} + \frac{1}{2}\hat{P}_{F} + \lambda_{P} + O\left(\|\xi\|^{3}\right), \quad \hat{P}^{*} = \frac{1}{2}\hat{P}_{H}^{*} + \frac{1}{2}\hat{P}_{F}^{*} + \lambda_{P^{*}} + O\left(\|\xi\|^{3}\right) \quad (88)$$

where

$$\lambda_P = \frac{1}{8} \left(\hat{P}_H - \hat{P}_F \right)^2, \quad \lambda_{P^*} = \frac{1}{8} \left(\hat{P}_H^* - \hat{P}_F^* \right)^2$$

And the second-order approximations for the price setting conditions are

$$\hat{P}_H = E\left[\hat{K} + \hat{P} + \rho\hat{C}\right] + \lambda_{P_H} + O\left(\|\xi\|^3\right)$$
(89)

$$\hat{P}_{F}^{*} = E\left[\hat{K}^{*} + \hat{P}^{*} + \rho\hat{C}^{*}\right] + \lambda_{P_{F}^{*}} + O\left(\|\xi\|^{3}\right)$$
(90)

where

$$\lambda_{P_H} = \frac{1}{2} E \left[\hat{K}^2 + 2\hat{K}\hat{Y} - \hat{P}^2 - \rho^2 \hat{C}^2 + \hat{Y}\hat{P} + \rho\hat{Y}\hat{C} - \rho\hat{P}\hat{C} \right]$$
$$\lambda_{P_F^*} = \frac{1}{2} E \left[\hat{K}^{*2} + 2\hat{K}^*\hat{Y}^* - \hat{P}^{*2} - \rho^2\hat{C}^{*2} + \hat{Y}^*\hat{P}^* + \rho\hat{Y}^*\hat{C}^* - \rho\hat{P}^*\hat{C}^* \right]$$

Notice that second-order terms are collected in the six terms λ_Y , λ_{Y^*} , λ_P , λ_{P^*} , λ_{P_H} and $\lambda_{P_F^*}$. Using the above equations it is possible to solve for the first moments of all the variables of the model in terms of these second-order terms. In this way the following expression is obtained for the first-order terms in the home welfare function

$$E\left[\hat{C} - \hat{Y}\right] = \frac{1}{2\left[1 + \rho\left(\theta - 1\right)\right]} E\left\{\lambda_{P_{H}} - \lambda_{P_{F}^{*}} - 2\left[1 + \rho\left(\theta - 1\right)\right]\lambda_{Y} + (1 - 2\theta)\left(1 + \rho\theta\right)\lambda_{P} - (1 - \rho\theta)\lambda_{P^{*}}\right\} + O\left(\left\|\xi\right\|^{3}\right) \quad (91)$$

Notice now that welfare can be written entirely in terms of second moments. The remaining task is therefore to derive expressions for the second moments of the variables of the model. This task is made easier by noting that second-order accurate solutions for second moments can be derived from first-order accurate solutions for the realisations of variables. First-order accurate solutions for *ex post* realisations can be obtained from equations (82) to (89) by ignoring second-order terms. In the case where $\rho = 1$ the resulting set of equations can be used to derive the following expressions for the λs

$$E\left[\lambda_{P_H}\right] = \frac{1}{2}E\left[\left(\hat{Y} + \hat{K}\right)^2\right] + O\left(\|\xi\|^3\right)$$
(92)

$$E\left[\lambda_{P_F^*}\right] = \frac{1}{2}E\left[\left(\hat{Y}^* + \hat{K}^*\right)^2\right] + O\left(\left\|\xi\right\|^3\right)$$
(93)

$$E[\lambda_{Y}] = E[\lambda_{Y^{*}}] = \frac{1}{8} (1-\theta)^{2} E[\hat{S}^{2}] + O(||\xi||^{3})$$
(94)

$$E\left[\lambda_P\right] = E\left[\lambda_{P^*}\right] = \frac{1}{8}\left(1-\theta\right)E\left[\hat{S}^2\right] + O\left(\left\|\xi\right\|^3\right)$$
(95)

Home welfare can therefore be written as follows

$$\tilde{\Omega} = -\frac{1}{4}E\left[\frac{1}{\theta}\left(2\theta - 1\right)\left(\hat{Y} + \hat{K}\right)^2 + \frac{1}{\theta}\left(\hat{Y}^* + \hat{K}^*\right)^2 + \frac{1}{2}\left(1 - \theta\right)\hat{S}^2\right] + O\left(\|\xi\|^3\right) \tag{96}$$

which is the expression used in the main text. The expression for foreign welfare follows immediately by symmetry.

The procedure for deriving welfare expressions for the risk-sharing case is identical. The only amendment required is to replace the current account equation with the risk-sharing condition $\hat{C} = \hat{C}^*$. In the case where risk sharing takes place after monetary rules are chosen (i.e. the case considered in the main text) the risk sharing equation is only relevant when deriving solutions for the *ex post* realisations of variables. It therefore only affects the second moments of variables. The current account condition continues to be relevant for the derivation of first-moment terms. But in the case where risk sharing takes place before monetary rules are chosen (i.e. the case considered immediately below in this Appendix) the risk sharing condition is imposed for the derivation of both first and second moment terms.

An Alternative Risk Sharing Structure

This appendix briefly considers the case where risk sharing takes place before monetary policy rules are determined. Only the case where $\rho = 1$ is considered. In this case the solution procedure described above can be used to show that home and foreign welfare can be written as follows

$$\tilde{\Omega} = -\frac{1}{4}E\left[\left(2-\theta\right)\left(\hat{Y}+\hat{K}\right)^2 + \theta\left(\hat{Y}^*+\hat{K}^*\right)^2 + 2\theta\left(1-\theta\right)\hat{S}^2\right]$$
(97)

and

$$\tilde{\Omega}^* = -\frac{1}{4}E\left[(2-\theta) \left(\hat{Y}^* + \hat{K}^* \right)^2 + \theta \left(\hat{Y} + \hat{K} \right)^2 + 2\theta \left(1 - \theta \right) \hat{S}^2 \right]$$
(98)

while output levels and the exchange rate are given by (49) and (50) in the main text.

Equations (97) and (98) show that the change in the timing of risk trading has a significant effect on the structure of the welfare function. It is still the case that welfare depends on the volatility of the disutility of work effort in both countries and the volatility of the exchange rate. But now there is a much stronger imbalance between the effects of the volatility of work effort at home and abroad. Consider the home country welfare function. For values of θ greater than 2 home welfare is increasing in the volatility of the disutility of home work effort and decreasing in the volatility of the disutility of foreign work effort. This is as a direct result of the change in timing of risk trading. The home policymaker now knows that home agents are 'insured' against any change in the expected level of their work effort. If the home policymaker chooses a monetary rule which increases the volatility of

θ	1	2	4	6	8	10
$ ilde{\Omega}_R^C - ilde{\Omega}_R^M$		0.750	1.250	1.750	2.250	2.750
$\tilde{\Omega}_R^N{-}\tilde{\Omega}_R^M$		-	-0.015	0.122	0.166	0.187
$\tilde{\Omega}_{R}^{C} {-} \tilde{\Omega}_{R}^{N}$	0.000	-	1.266	1.628	2.084	2.563
$\frac{\frac{100 \times \left(\tilde{\Omega}_{R}^{C} - \tilde{\Omega}_{R}^{N}\right)}{\left(\tilde{\Omega}_{R}^{C} - \tilde{\Omega}_{R}^{M}\right)}$	0.0	-	101.2	93.0	92.6	93.2

Table 8: The welfare effects of coordination: Alternative risk-sharing structure

the disutility of home work effort this will increase home goods prices and reduce home work effort. Home agents will have a lower expected level of income but their consumption level will be tied to the expected level of world output by the risk-sharing arrangement. Home agents therefore benefit from an increase in leisure time while receiving the world average level of consumption. In other words the home policymaker believes that is possible to shift the burden of production onto the foreign economy. This mechanism clearly creates a further spillover effect of monetary policy which potentially increases the gains from monetary policy coordination.

The quantitative implications of the additional spillover effects arising in this case are illustrated in Table 8. The parameter values and construction of the table are identical to the cases discussed in the main text. It is clear that the gains from coordination can now be very large, both in relative and absolute terms. When compared to the case where risk trading takes place after policy rules are chosen non-coordinated policymaking now yields much lower levels of welfare.

References

- Benigno, Gianluca and Pierpaolo Benigno (2001a) "Price Stability as a Nash Equilibrium in Monetary Open Economy Models" CEPR Discussion Paper No 2757.
- [2] Benigno, Gianluca and Pierpaolo Benigno (2001b) "Implementing Monetary Cooperation through Inflation Targeting" unpublished manuscript, Bank of England and New York University.
- Benigno, Pierpaolo (2001) "Price Stability with Imperfect Financial Integration" CEPR Discussion Paper No 2854.
- [4] Calvo, Guillermo A (1983) "Staggered Prices in a Utility-Maximising Framework" Journal of Monetary Economics, 12, 383-398.

- [5] Canzoneri, Matthew B and Dale W Henderson (1991) Monetary Policy in Interdependent Economies: A Game Theoretic Approach MIT Press, Cambridge MA.
- [6] Clarida, Richard H, Jordi Gali and Mark Gertler (2001) "A Simple Framework for International Monetary Policy Analysis" unpublished manuscript, Columbia University, Pompeu-Fabra and New York University.
- [7] Cole, Harold and Maurice Obstfeld (1991) "Commodity Trade and International Risk Sharing: How Much do Financial Markets Matter?" Journal of Monetary Economics, 28, 3-24.
- [8] Corsetti, Giancarlo and Paolo Pesenti (2001a) "Welfare and Macroeconomic Interdependence" Quarterly Journal of Economics, 116, 421-446.
- [9] Corsetti, Giancarlo and Paolo Pesenti (2001b) "International Dimensions of Optimal Monetary Policy" NBER Working Paper No 8230.
- [10] Currie, David and Paul Levine (1984) "Macroeconomic Policy Design in an Interdependent World" in Buiter, Willem and Richard Marston (eds) International Economic Policy Coordination, Cambridge University Press, Cambridge.
- [11] Devereux, Michael B and Charles Engel (1998) "Fixed versus Floating Exchange Rates: How Price Setting Affects the Optimal Choice of Exchange Rate Regime" NBER Working paper 6867.
- [12] Devereux, Michael B and Charles Engel (2000) "Monetary Policy in an Open Economy Revisited: Price Setting and Exchange Rate Flexibility" NBER Working Paper No 7665.
- [13] Devereux, Michael B (2001) "International Financial Markets and the Exchange Rate: Re-evaluating the Case for Flexible Exchange Rates" unpublished manuscript.
- [14] Goodfriend, Marvin and Robert G King (2001) "The Case for Price Stability" NBER Working Paper No 8423.
- [15] Kehoe, Patrick J and Fabrizio Perri (2000) "International Business Cycles with Endogenous Incomplete Markets" NBER Working Paper No 7870.
- [16] King, Robert G and Alexander L Wolman (1999) "What Should the Monetary Authority Do When Prices are Sticky?" in John B Taylor (ed) Monetary Policy Rules, University of Chicago Press, Chicago.

- [17] Kim, Jinill and Sunghyun H Kim (2000) "Spurious Welfare Reversals in International Business Cycle Models" unpublished manuscript, University of Virginia (forthcoming in the Journal of International Economics).
- [18] Lane, Philip (2001) "The New Open Economy Macroeconomics: A Survey" Journal of International Economics, 54-235-266.
- [19] Ligon, Ethan, Jonathan P Thomas and Tim Worrall (1997) "Informal Insurance Arrangements in Village Economies" unpublished manuscript, University of St Andrews.
- [20] Ligon, Ethan, Jonathan P Thomas and Tim Worrall (2000) "Mutual Insurance, Individual Savings and Limited Commitment" *Review of Economic Dynamics*, 3, 216-246.
- [21] Miller, Marcus and Mark Salmon (1984) "Policy Coordination and Dynamic Games" in Buiter, Willem and Richard Marston (eds) International Economic Policy Coordination, Cambridge University Press, Cambridge.
- [22] Obstfeld, Maurice and Kenneth Rogoff (1995) "Exchange Rate Dynamics Redux" Journal of Political Economy, 103, 624-660.
- [23] Obstfeld, Maurice and Kenneth Rogoff (1998) "Risk and Exchange Rates" NBER Working paper 6694.
- [24] Obstfeld, Maurice and Kenneth Rogoff (2002) "Global Implications of Self-Oriented National Monetary Rules" *Quarterly Journal of Economics*, 117, 503-536.
- [25] Oudiz, Gilles and Jeffrey Sachs (1984) "Macroeconomic Policy Coordination among the Industrial Countries" Brookings Papers on Economic Activity, 1, 1-64.
- [26] Rogoff, Kenneth (1985) "Can International Monetary Cooperation be Counterproductive?" Journal of International Economics, 18, 199-217.
- [27] Schmitt-Grohé, Stephanie and Martin Uribe (2001) "Solving Dynamic General Equilibrium Models using a Second-Order Approximation to the Policy Function" CEPR Discussion Paper No 2963.
- [28] Sims, Christopher (2000) "Second-Order Accurate Solutions of Discrete Time Dynamic Equilibrium Models" unpublished manuscript, Princeton University.

- [29] Sutherland, Alan (2001) "A Simple Second-Order Solution Method for Dynamic General Equilibrium Models" unpublished manuscript, University of St Andrews.
- [30] Sutherland, Alan (2002) "The Expenditure Switching Effect, Welfare and Monetary Policy in an Open Economy" unpublished manuscript, University of St Andrews.
- [31] Svensson, Lars E O (1999) "Inflation Targeting as a Monetary Policy Rule" Journal of Monetary Economics, 43, 607-654.
- [32] Tille, Cedric (1999) "The Role of Consumption Substitutability in the International Transmission of Shocks" forthcoming in the *Journal of International Economics*.
- [33] van Wincoop, Eric (1994) "Welfare Gains from International Risk Sharing" Journal of Monetary Economics, 34, 175-200.
- [34] Woodford, Michael (2001) "Inflation Stabilisation and Welfare" NBER Working Paper No 8071.

European Central Bank working paper series

For a complete list of Working Papers published by the ECB, please visit the ECB's website (http://www.ecb.int).

- 113 "Financial frictions and the monetary transmission mechanism: theory, evidence and policy implications" by C. Bean, J. Larsen and K. Nikolov, January 2002.
- 114 "Monetary transmission in the euro area: where do we stand?" by I. Angeloni, A. Kashyap, B. Mojon, D. Terlizzese, January 2002.
- 115 "Monetary policy rules, macroeconomic stability and inflation: a view from the trenches" by A. Orphanides, December 2001.
- 116 "Rent indices for housing in West Germany 1985 to 1998" by J. Hoffmann and C. Kurz., January 2002.
- 117 "Hedonic house prices without characteristics: the case of new multiunit housing" by O. Bover and P. Velilla, January 2002.
- 118 "Durable goods, price indexes and quality change: an application to automobile prices in Italy, 1988-1998" by G. M. Tomat, January 2002.
- 119 "Monetary policy and the stock market in the euro area" by N. Cassola and C. Morana, January 2002.
- 120 "Learning stability in economics with heterogenous agents" by S. Honkapohja and K. Mitra, January 2002.
- 121 "Natural rate doubts" by A. Beyer and R. E. A. Farmer, February 2002.
- 122 "New technologies and productivity growth in the euro area" by F. Vijselaar and R. Albers, February 2002.
- 123 "Analysing and combining multiple credit assessments of financial institutions" by E. Tabakis and A. Vinci, February 2002.
- 124 "Monetary policy, expectations and commitment" by G. W. Evans and S. Honkapohja, February 2002.
- 125 "Duration, volume and volatility impact of trades" by S. Manganelli, February 2002.
- 126 "Optimal contracts in a dynamic costly state verification model" by C. Monnet and E. Quintin, February 2002.
- 127 "Performance of monetary policy with internal central bank forecasting" by S. Honkapohja and K. Mitra, February 2002.
- 128 "Openness, imperfect exchange rate pass-through and monetary policy" by F. Smets and R. Wouters, February 2002.

- 129 "Non-standard central bank loss functions, skewed risks, and certainty equivalence" by A. al-Nowaihi and L. Stracca, March 2002.
- 130 "Harmonized indexes of consumer prices: their conceptual foundations" by E. Diewert, March 2002.
- 131 "Measurement bias in the HICP: what do we know, and what do we need to know?" by M. A. Wynne and D. Rodríguez-Palenzuela, March 2002.
- 132 "Inflation dynamics and dual inflation in accession countries: a "new Keynesian" perspective" by O. Arratibel, D. Rodríguez-Palenzuela and C. Thimann, March 2002.
- 133 "Can confidence indicators be useful to predict short term real GDP growth?" by A. Mourougane and M. Roma, March 2002.
- 134 "The cost of private transportation in the Netherlands, 1992-1999" by B. Bode and J. Van Dalen, March 2002.
- 135 "The optimal mix of taxes on money, consumption and income" by F. De Fiore and P. Teles, April 2002.
- 136 "Retail bank interest rate pass-through: the new evidence at the euro area level" by G. de Bondt, April 2002.
- 137 "Equilibrium bidding in the eurosystem's open market operations" by U. Bindseil, April 2002.
- 138 "New" views on the optimum currency area theory: what is EMU telling us?" by F. P. Mongelli, April 2002.
- 139 "On currency crises and contagion" by M. Fratzscher, April 2002.
- 140 "Price setting and the steady-state effects of inflation" by M. Casares, May 2002.
- 141 "Asset prices and fiscal balances" by F. Eschenbach and L. Schuknecht, May 2002.
- 142 "Modelling the daily banknotes in circulation in the context of the liquidity management of the European Central Bank", by A. Cabrero, G. Camba-Mendez, A. Hirsch and F. Nieto, May 2002.
- 143 "A non-parametric method for valuing new goods", by I. Crawford, May 2002.
- 144 "A failure in the measurement of inflation: results from a hedonic and matched experiment using scanner data", by M. Silver and S. Heravi, May 2002.
- 145 "Towards a new early warning system of financial crises", by M. Fratzscher and M. Bussiere, May 2002.
- 146 "Competition and stability what's special about banking?", by E. Carletti and P. Hartmann, May 2002.

- 147 "Time-to-build approach in a sticky price, stricky wage optimizing monetary model, by M. Casares, May 2002.
- 148 "The functional form of yield curves" by V. Brousseau, May 2002.
- 149 "The Spanish block of the ESCB-multi-country model" by A. Estrada and A. Willman, May 2002.
- 150 "Equity and bond market signals as leading indicators of bank fragility" by R. Gropp, J. Vesala and G. Vulpes, June 2002.
- 151 "G-7 inflation forecasts" by F. Canova, June 2002.
- 152 "Short-term monitoring of fiscal policy discipline" by G. Camba-Mendez and A. Lamo, June 2002.
- 153 "Euro area production function and potential output: a supply side system approach" by A. Willman, June 2002.
- 154 "The euro bloc, the dollar bloc and the yen bloc: how much monetary policy independence can exchange rate flexibility buy in an interdependent world?" by M. Fratzscher, June 2002.
- 155 "Youth unemployment in the OECD: demographic shifts, labour market institutions, and macroeconomic shocks" by J. F. Jimeno and D. Rodriguez-Palenzuela, June 2002.
- 156 "Identifying endogenous fiscal policy rules for macroeconomic models" by J. J. Perez, and P. Hiebert, July 2002.
- 157 "Bidding and performance in repo auctions: evidence from ECB open market operations" by K. G. Nyborg, U. Bindseil and I. A. Strebulaev, July 2002.
- 158 "Quantifying Embodied Technological Change" by P. Sakellaris and D. J. Wilson, July 2002.
- 159 "Optimal public money" by C. Monnet, July 2002.
- 160 "Model uncertainty and the equilibrium value of the real effective euro exchange rate" by C. Detken, A. Dieppe, J. Henry, C. Marin and F. Smets, July 2002.
- 161 "The optimal allocation of risks under prospect theory" by L. Stracca, July 2002.
- 162 "Public debt asymmetries: the effect on taxes and spending in the European Union" by S. Krogstrup, August 2002.
- 163 "The rationality of consumers' inflation expectations: survey-based evidence for the euro area" by M. Forsells and G. Kenny, August 2002.
- 164 "Euro area corporate debt securities market: first empirical evidence" by G. de Bondt, August 2002.

- 165 "The industry effects of monetary policy in the euro area" by G. Peersman and F. Smets, August 2002.
- 166 "Monetary and fiscal policy interactions in a micro-founded model of a monetary union" by R. M.W.J. Beetsma and H. Jensen, August 2002.
- 167 "Identifying the effects of monetary policy shocks on exchange rates using high frequency data" by J. Faust, J.H. Rogers, E. Swanson and J.H. Wright, August 2002.
- 168 "Estimating the effects of fiscal policy in OECD countries" by R. Perotti, August 2002.
- 169 "Modeling model uncertainty" by A. Onatski and N. Williams, August 2002.
- 170 "What measure of inflation should a central bank target?" by G. Mankiw and R. Reis, August 2002.
- 171 "An estimated stochastic dynamic general equilibrium model of the euro area" by F. Smets and R. Wouters, August 2002.
- 172 "Constructing quality-adjusted price indices: a comparison of hedonic and discrete choice models" by N. Jonker, September 2002.
- 173 "Openness and equilibrium determinacy under interest rate rules" by F. de Fiore and Z. Liu, September 2002.
- 174 "International monetary policy coordination and financial market integration" by A. Sutherland, September 2002.